[#BloombergLP-bdlt-DayOfWeek-Enum]
= xref:BloombergLP.adoc[BloombergLP]::xref:BloombergLP/bdlt.adoc[bdlt]::xref:BloombergLP/bdlt/DayOfWeek.adoc[DayOfWeek]::Enum
:relfileprefix: ../../../
:mrdocs:

Enumerated day&hyphen;of&hyphen;week values&period;

== Synopsis

Declared in `&lt;bdlt&lowbar;dayofweek&period;h&gt;`

[source,cpp,subs="verbatim,replacements,macros,-callouts"]
----
enum Enum;
----

== Members

[cols="1,4"]
|===
| Name| Description
| `e&lowbar;SUN` 
| Sunday&period;
| `e&lowbar;SUNDAY` 
| Alias for `e&lowbar;SUN`&period;
| `e&lowbar;MON` 
| Monday&period;
| `e&lowbar;MONDAY` 
| Alias for `e&lowbar;MON`&period;
| `e&lowbar;TUE` 
| Tuesday&period;
| `e&lowbar;TUESDAY` 
| Alias for `e&lowbar;TUE`&period;
| `e&lowbar;WED` 
| Wednesday&period;
| `e&lowbar;WEDNESDAY` 
| Alias for `e&lowbar;WED`&period;
| `e&lowbar;THU` 
| Thursday&period;
| `e&lowbar;THURSDAY` 
| Alias for `e&lowbar;THU`&period;
| `e&lowbar;FRI` 
| Friday&period;
| `e&lowbar;FRIDAY` 
| Alias for `e&lowbar;FRI`&period;
| `e&lowbar;SAT` 
| Saturday&period;
| `e&lowbar;SATURDAY` 
| Alias for `e&lowbar;SAT`&period;
| `SUN` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `MON` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `TUE` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `WED` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `THU` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `FRI` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `SAT` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `SUNDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `MONDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `TUESDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `WEDNESDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `THURSDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `FRIDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `SATURDAY` 
| Deprecated three&hyphen;letter aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;SUN` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;SUNDAY` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;MON` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;TUE` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;TUESDAY` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;WED` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;THU` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;THURSDAY` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;FRI` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;FRIDAY` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;SAT` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
| `BDET&lowbar;SATURDAY` 
| Deprecated `BDET&lowbar;&ast;` aliases for the day&hyphen;of&hyphen;week enumerators&period;
|===


== Non-Member Functions

[cols="1,4"]
|===
| Name| Description
| xref:BloombergLP/bdex_VersionFunctions/maxSupportedVersion-09.adoc[`&colon;&colon;BloombergLP&colon;&colon;bdex&lowbar;VersionFunctions&colon;&colon;maxSupportedVersion`]
| Return the maximum supported BDEX version for `DayOfWeek&colon;&colon;Enum`&period;
| xref:BloombergLP/bdlt/maxSupportedBdexVersion-077.adoc[`&colon;&colon;BloombergLP&colon;&colon;bdlt&colon;&colon;maxSupportedBdexVersion`]
| Return the maximum valid BDEX format version for this type&period;
|===


[.small]#Created with https://www.mrdocs.com[MrDocs]#
